Dear Editor,

Thank you for carefully reviewing our manuscript previously titled “Type A aortic dissection following heart transplantation: a case report” for possible publication in the *World Journal of Clinical Cases*. We are grateful to you and your reviewers for their constructive critique. We have revised the manuscript, highlighting our revisions in red, and have attached point-by-point responses detailing how we have revised the manuscript in response to the reviewers' comments below.

Thank you for your consideration and further review of our manuscript. Please do not hesitate to contact us with any further questions or recommendations.

Yours Sincerely,

Zhu Zeng
Reviewer Comments:

Reviewer #1:

Response:

The history of heart transplantation, laboratory examination and medications have been modified. The side effects of anti-rejection medications may be the cause of obesity and hypertension. The error of the vagus artery has been corrected. Clinical opinion are not expressed because of the word limit of the article.

Reviewer #2:

Response:

This case was approved by the Medical Ethics Committee of Union Hospital affiliated to Tongji Medical College of Huazhong University of Science and Technology and consent inform have been acquired. The patient went to the nearest hospital and didn't come to our hospital for further consultation.